Author's response to reviews

Title: Bone preserving level of osteotomy in short-stem total hip arthroplasty does not influence stress shielding dimensions - a comparing finite elements analysis

Authors:

Rene Burchard (dr_burchard@t-online.de)
Sabrina Braas (sabrina.braas@gmx.de)
Christian Soost (christian.soost@uni-siegen.de)
Jan Graw (ja.graw@web.de)
Jan Schmitt (janschmitt@arcor.de)

Version: 2 Date: 27 Jul 2017

Author’s response to reviews:

Dear Faye Nourollahi,

Thank you very much for the opportunity to resubmit our manuscript for consideration and publication in BMC Musculoskeletal Disorders.

We have revised our paper following your suggestions:

1. Please pick a final title for your manuscript. Please only list that title in the submission system and in the main manuscript.

We have change the title.

2. Thank you for submitting an Ethics section. Please clarify in this section that all participates provided informed consent to participate.

Informed consent to participate is declared in the methods section on page 7 and in a additional section on page 13.

3. We would appreciate if you added a section titled “Supplementary files” where you list the following information about your supplementary material:

- File name
- Title of data
- Description of data

Please ensure that all additional files are explicitly referred to in the main text. Any items which do not meet these requirements may be deleted by our production department.

We have implement it.

4. Please delete supplementary files, "Revision_Coverletter_BMSD-D-17-00428.doc" and "BMSD-D-17-00428_point-by-point-response.docx". They are no longer necessary at this stage of production.

We have delete it.

5. Upon uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

We have implement it.

We hope that the manuscript now adheres to the Journal’s standards and expectations. If questions remain please do not hesitate to contact us.

Yours sincerely,

Dr. Rene Burchard
On behalf of the authors